Author's response to reviews

Title: Consumption of sweet foods and mammographic breast density

Authors:

Caroline Duchaine (caroline.duchaine.1@ulaval.ca)
Isabelle Dumas (isabelle.dumas.4@ulaval.ca)
Caroline Diorio (caroline.diorio@uresp.ulaval.ca)

Version: 2
Date: 17 December 2013

Author's response to reviews: see over
Editor-in-chief, *BMC Public Health*

**Object:** Manuscript entitled “Consumption of sweet foods and mammographic breast density”

Dear Editor,

We are pleased to submit to your attention the following manuscript, entitled “Consumption of sweet foods and mammographic breast density” for consideration to be published in *BMC Public Health*.

To our knowledge, this is the first article to suggest that higher intake of sugar-sweetened beverages is associated to higher mammographic breast density among adult women. Considering the high consumption of these drinks in the general population and the strength of the association between mammographic density and breast cancer risk, this finding could open new avenues for breast cancer prevention.

All of the authors have read and approved the paper and it has not been published previously nor is being considered by any other peer-reviewed journal. None of the authors had a personal or financial conflict of interest.

Thank you for receiving our manuscript and considering it for publication. We appreciate your time and look forward to hear from you.

Caroline Diorio, PhD  
Axe oncologie  
Centre de recherche du CHU de Québec  
1050 Chemin Sainte-Foy  
Quebec City, QC, Canada G1S 4L8  
Tel: (418) 682-7390  
Fax: (418) 682-7949  
E-mail: caroline.diorio@uresp.ulaval.ca